Reviewer #1:
Scientific Quality: Grade B (Very good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Accept (General priority)
Specific Comments to Authors: In this paper, the author made a scientometric analysis of the articles published in World Journal of Clinical Cases in the past 5 years and presented the data to the readers. The results show that the journal has been growing with time with appropriate publication standards as per the COPE guidelines. I think that this paper is well-written and very interesting. This paper is worth publishing in World Journal of Clinical Cases.
Response: Thanks for the encouraging words.

Reviewer #2:
Scientific Quality: Grade A (Excellent)
Language Quality: Grade A (Priority publishing)
Conclusion: Accept (High priority)
Specific Comments to Authors: The journal has evolved over time and has established appropriate publication standards according to COPE guidelines. The journal has a wide range of published articles covering all areas of medicine to meet the aims and scope of the journal.
Response: Thanks for the encouraging words.